Author's response to reviews

Title: Comparison of the efficacy and safety of URSL, RPLU, and MPCNL for treatment of large upper impacted ureteral stones: A Randomized Controlled Trial

Authors:

Yunyan Wang (yunyanwang2000@163.com)
Bing Zhong (15152569186@163.com)
Xiaosong Yang (haxiaosong@126.com)
Gongcheng Wang (wgc1955@sina.com)
Peijin Hou (hyhou@163.com)
Junsong Meng (haywyy1322@163.com; yyyh622@126.com)

Version: 3 Date: 12 Jun 2017

Author’s response to reviews:

Manuscript ID: BURO-D-16-00369R2

Title: Comparison of the efficacy and safety of URSL, RPLU, and MPCNL for treatment of large upper impacted ureteral stones: A Randomized Controlled Trial

Journal: BMC Urology

Response to Reviewers' comments

Dear editors,

We thank you for your careful consideration of our manuscript. We appreciate your response and overall positive initial feedback, and made modifications to improve the manuscript. After carefully reviewing the comments made by the Reviewers, we have modified the manuscript to improve the presentation of our results and their discussion, therefore providing a more complete context for the research that may be of interest to your readers.

We hope that you will find the revised paper suitable for publication, and we look forward to contributing to your journal. Please do not hesitate to contact us with other questions or concerns regarding the manuscript.
Best regards,

Junsong Meng
Department of Urology, Huai’an First People’s Hospital, Nanjing Medical University, No.6 Beijing West Road, Huai’an 223300, Jiangsu, China
Tel: +86-13505239046
E-mail: haywyyy1322@163.com

Thank you for providing a section on Ethics approval. We would appreciate it if you could please provide us with the full, official name of the committee which granted you approval for this study, and committee reference number if appropriate.

Response: We have added the full name of the Ethic Committee and the committee number.